Author's response to reviews

Title: Effect of community-based health management on the elderly health: a prospective randomized control trial from China

Authors:

Jianqian Chao (chaoseu@163.com)
Yimin Wang (89wangyimin@sina.com)
Hui Xu (ghyyxh@yahoo.com)
Qing Yu (Yuqing8886@sina.com)
Lili Jiang (lily.jiang.cn@163.com)
Lin Tian (tiandarling@163.com)
Wenyuan Xie (wenyuan218@126.com)
Pei Liu (liupeiseu@126.com)

Version: 2 Date: 29 October 2011

Author's response to reviews:

Dear Editor,

First I thank for your suggestion. I have added "Ethical approval" in the Methods section. It includes "The Medical Ethics Committee of Public Health School in Southeast University has granted ethical approval. The trial is registered as IRB00001760." Because the abstract has been the trial registration number, I do not change it.